Author's response to reviews

Title: Case report: Scleroderma with crescentic glomerulonephritis

Authors:

Arunachalam Ramaswami (ramsdr@yahoo.com)
Thiraviam Kandaswamy (tknephro@yahoo.com)
Tholappan Rajendran (ammukkannu@yahoo.com)
kizhakkePisharam Jeyakrishnan (drjayak@yahoo.com)
Hla Aung (hlaaung@brunet.bn)
Mohammad lqbal (IQBAL103@hotmail.com)
Chakko K Jacob (ckjacob@hotmail.co.uk)
Shaukat Zinna (zinnashaukat@yahoo.com)
Gazala Kafeel (gbilgrami@hotmail.com)

Version: 6 Date: 14 December 2007

Author's response to reviews:

Reply 1: Text is altered suitably. 2. Text is altered suitably. 3. Histopathology is interpreted as immune complex mediated disease. We don't consider lupus nephritis as underlying disease here because C3 C4 levels are not low and the immunoperoxide staining did not reveal a "full house" immunoglobulin deposits. Patient did not have clinical features of SLE. 4. Patient was negative for 3rd generation anti HCV antibodies. RA factor level was 8 IU /ml and did not have abnormal liver enzyme levels.